Author's response to reviews

Title: Toxic risk of stereotactic body radiotherapy and concurrent helical tomotherapy followed by erlotinib for non-small-cell lung cancer treatment - case report

Authors:

Chen-Hsi Hsieh (chenci28@ms49.hinet.net)
Hou-Tai Chang (houtai38@gmail.com)
Shih-Chiang Lin (pigchiang@yahoo.com.tw)
Yu-Jen Chen (chenmdphd@yahoo.com)
Li-Ying Wang (living@ntu.edu.tw)
Yen-Ping Hsieh (fannyhsieh@hotmail.com)
Chien-An Chen (kenk102000@yahoo.com.tw)
Ngot-Swan Chong (nschong@mail.femh.org.tw)
Shoei Long Lin (canceronline@gmail.com)
Chun-Yi Chen (canceronline@gmail.com)
Pei-Wei Shueng (shueng@hotmail.com)

Version: 4 Date: 23 December 2010

Author's response to reviews: see over
Professor Melissa Norton,  

December 23, 2010

Editor-in-Chief,  

*BMC Cancer*

Dear Professor Norton:

On behalf of all authors, I appreciate the time and effort of the editor and reviewers in critiquing our work (MS: 3100979884018932). Attached is a point-by-point response to the Reviewers’ comments. We submit this manuscript for reconsideration for publication in *BMC Cancer*. Thank you for your great effort on our work.

Yours sincerely,

Pei-Wei Shueng, MD  
Department of Radiation Oncology,  
Far Eastern Memorial Hospital,  
21, Nan-Ya South Road, Section 2 Pan-Chiao, Taipei 220, Taiwan  
Fax: 886-2-8966-0906  
Phone: 886-2-8966-7000 ext. 1033  
E-mail: shueng@hotmail.com
Major revisions (we require the author to make these changes)

Comment: Title: please ensure that the title is identical in the manuscript and the submission system. Currently they do not match.
Response: We revised accordingly.

Comment: Abstract: please ensure that the Abstract is present in the manuscript and matches exactly what is in the submission system. Currently they do not match.
Response: We revised accordingly.

Comment: Figures: You have cited and provided legends for 2 figures, but you have uploaded 6 figures. Please note that each file uploaded is automatically denoted as a new figure. For example, if you make 3A and 3B separate files, they will come up as figure 3 and figure 4. Please renumber the figures and adjust citations accordingly. Alternatively, you can place all parts of a figure (e.g. A-D) into a single-page composite figure file.
Response: We renumber the figures and adjust citations accordingly.

Comment: References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].' Currently reference 15 is cited before reference 14 in the manuscript. Please ensure that all references are cited in consecutive numerical order in the manuscript.
Response: We recited the references and adjust citations in consecutive numerical order in the manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Comment: Line numbers: Please remove these.
Response: The line numbers have been removed.

Comment: Text: Please ensure that all text is displayed in black throughout the manuscript.
Response: All text is displayed in black throughout the manuscript accordingly.
Comment: Figures: Please ensure all mention of figures is consistent (it should be Figure or Fig but not a mixture of both).

Response: All mention of figures is consistent in the manuscript.

Comment: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: The manuscript has been check.